Author's response to reviews

Title: Drug use and associated factors among pregnant mothers attending antenatal care at governmental health centers in Bahir Dar city administration, Northwest Ethiopia, cross sectional study.

Authors:

Chanie Admasie (chaneadmasse@gmail.com)
Belaynew Wassie (bewassie@gmail.com)
Gedefaw Abeje Fekadu (abejegedefaw@gmail.com)

Version: 2 Date: 28 July 2014

Author's response to reviews:

Dear Editor, We authors of this manuscript have made many changes on the manuscript. The details of the change are highlighted yellow in the main manuscript.

We have also made point by point response to the reviewers comments. I have attached the point by point response and the revised manuscript. If more comments, we are ready to manage and make corrections.